Author's response to reviews

Title: Association between ratio of measured extracellular volume to expected body fluid volume and renal outcomes in patients with chronic kidney disease: a retrospective single-center cohort study

Authors:

Reibin Tai (rebin_1027@med.toho-u.ac.jp)
Yasushi Ohashi (ghashiy@med.toho-u.ac.jp)
Sonoo Mizuiri (sm210@med.toho-u.ac.jp)
Atsushi Aikawa (aaikawa@med.toho-u.ac.jp)
Ken Sakai (kensakai@med.toho-u.ac.jp)

Version: 4 Date: 14 November 2014

Author's response to reviews: see over
Dear Dr. Liffert Vogt,

Enclosed please find a revised version of our manuscript (MS: 9600796221302336) titled “Association between ratio of measured extracellular volume to expected body fluid volume and renal outcomes in patients with chronic kidney disease: a retrospective single-center cohort study” by Tai et al. We thank the editor for your careful analysis of our work. Your comments have led to alterations that have significantly enhanced the quality of our manuscript. A point-by-point response to each comment is outlined below.

Responses to editor’s comments:

In accordance with Editor’s request, we omitted “(e.g., the Watson formula)” in the line 84 and added “In addition” in the line 90 of the revised manuscript.

The manuscript has been carefully reviewed by an experienced editor of Edanz whose first language is English and who specializes in editing papers written by scientists whose native language is not English.

Thank you very much for considering publishing our work in *BMC Nephrology*. We really look forward to hearing from you.

Sincerely,

Yasushi Ohashi, MD